**Author's response to reviews**

**Title:** Early over expression of messenger RNA for multiple genes, including insulin, in the Pancreatic Lymph Nodes of NOD mice is associated with Islet Autoimmunity

**Authors:**

Beatrice Regnault (regnault@pasteur.fr)
Jose Osorio y Fortéa (Jose.OSORIOYFORTEA@limagrain.com)
Dongmei Miao (Dongmei.Miao@ucdenver.edu)
George S Eisenbarth (George.Eisenbarth@ucdenver.edu)
Evie Melanitou (evie.melanitou@pasteur.fr)

**Version:** 3  **Date:** 10 September 2009

**Author's response to reviews:**

Final proof revisions of the manuscript:

This last uploaded version corresponds to the formatting requirements that have been asked by the Editor. These requirements are as follows:

- Table shading has been removed and replaced by bold characters.
- Figure 3 has been corrected and submitted in only one page.
- Reference 55 has been completed.
- Titles of Tables have been placed on the top of the tables and additional text appears below as requested.
- Additional material has been renamed and conformed to the journal’s requirements.
- Figures have been cropped as much as possible.
- We have proof read all the text and some minor corrections have been made.